Author's response to reviews

Title: Induction of lung lesions in Wistar rats by 4-(methylnitrosamino)-1-(3-pyridyl)-1-butanone and their inhibitions by aspirin and phenethyl isothiocyanate.

Authors:

Bo Ye (ella_yb@hotmail.com)
Yu-Xia Zhang (zhangyuxiacn@hotmail.com)
Fei Yang (yangfei72@hotmail.com)
Hong-Lei Cheng (honglichen75@126.com)
Dong Xia (whxiadong@126.com)
Ming-Qiu Liu (xyxyi@126.com)
Bai-Tang Lai (baitanglai@yahoo.com.cn)

Version: 3 Date: 8 December 2006

Dear Editor:

I have finished my revised paper, and resubmitted online, the following is the point-by-point answers to the questions.

Reviewer: Katsumi Imaida
I am so sorry to bring you some trouble in reviewing my manuscript, because of my fault in writing. Intratracheal instillation were established and used in our department for 20 years. The first manuscript was published by Chinese Medical Journal in 1984 (reference 15). The treatment didn't change too much during these years, that is why we ignore the detail. We had added it in Materials and Methods.

I do appreciate reviewers' advices and thank for the hard works done by reviewers.

With best wishes
Bo Ye
Tel:86-027-87331241
E-mail: ella_yb@hotmail.com